

A Proteomic Investigation of Hepatic Resistance to *Ascaris* in a Murine Model

Gwendoline Deslyper, Thomas J. Colgan, Andrew J. R. Cooper, Celia V. Holland, James C. Carolan

Published: August 4, 2016 • https://doi.org/10.1371/journal.pntd.0004837

Abstract

The helminth Ascaris causes ascariasis in both humans and pigs. Humans, especially children, experience significant morbidity including respiratory complications, growth deficits and intestinal obstruction. Given that 800 million people worldwide are infected by Ascaris, this represents a significant global public health concern. The severity of the symptoms and associated morbidity are related to the parasite burden and not all hosts are infected equally. While the pathology of the disease has been extensively examined, our understanding of the molecular mechanisms underlying resistance and susceptibility to this nematode infection is poor. In order to investigate host differences associated with heavy and light parasite burden, an experimental murine model was developed utilising *Ascaris*-susceptible and -resistant mice strains, C57BL/6J and CBA/Ca, respectively, which experience differential burdens of migratory Ascaris larvae in the host lungs. Previous studies identified the liver as the site where this difference in susceptibility occurs. Using a label free quantitative proteomic approach, we analysed the hepatic proteomes of day four post infection C57BL/6J and CBA/Ca mice with and without Ascaris infection to identify proteins changes potentially linked to both resistance and susceptibility amongst the two strains, respectively. Over 3000 proteins were identified in total and clear intrinsic differences were elucidated between the two strains. These included a higher abundance of mitochondrial proteins, particularly those associated with the oxidative phosphorylation pathway and reactive oxygen species (ROS) production in the relatively resistant CBA/Ca mice. We hypothesise that the increased ROS levels associated with higher levels of mitochondrial activity results in a highly oxidative cellular environment that has a dramatic effect on the nematode's ability to successfully sustain a parasitic association with its resistant host. Under infection, both strains had increased abundances in proteins associated with the oxidative phosphorylation pathway, as well as the tricarboxylic acid cycle, with respect to their controls, indicating a general stress response to *Ascaris* infection. Despite the early stage of infection, some immune-associated proteins were identified to be differentially abundant, providing a novel insight into the host response to Ascaris. In general, the susceptible C57BL/6J mice displayed higher abundances in immune-associated proteins, most likely signifying a more active nematode cohort with respect to their CBA/Ca counterparts. The complement component C8a and S100 proteins, S100a8 and S100a9, were highly differentially abundant in both infected strains, signifying a potential innate immune response and the importance of the complement pathway in defence against macroparasite infection. In addition, the signatures of an early adaptive immune response were observed through the presence of proteins, such as plastin-2 and dipeptidyl peptidase 1. A marked decrease in proteins associated with translation was also observed in both C57BL/6J and CBA/Ca mice under infection, indicative of either a general response to Ascaris or a modulatory effect by the nematode itself. Our research provides novel insights into the in vivo host-Ascaris relationship on the molecular level and provides new research perspectives in the development of Ascaris control and treatment strategies

Author Summary

Ascaris infection is a significant burden on the people who live in developing countries with infection being linked to poor hygiene and low socio-economic status. The parasite causes a range of symptoms, especially in children, which include both chronic morbidity, such as growth retardation, and acute outcomes, such as intestinal obstruction. Certain people tend to be more heavily infected than others, with those individuals experiencing worse morbidity. The understanding of the difference between susceptible and resistant people is an essential first step in the development of new therapies in order to eliminate this neglected parasitic disease. Using an established mouse model involving a susceptible and resistant strain, we aimed to gain insight into the host-*Ascaris* interaction at the hepatic interface and elucidate some of the molecular mechanisms potentially involved in resistance. A number of key intrinsic differences were determined between both strains including major differences in mitochondrial and ROS associated processes which may present the nematodes with differing oxidative conditions and explain the failure of the nematode to establish a successful parasitism in the resistant strain. In addition, we resolved signatures of the innate and early adaptive immune response and a major reduction in the proteins associated with translation in both strains under infection. Our findings need to be further explored, but could be the foundation for a better understanding of the mechanisms behind the differential parasite burden and in the future, potential new therapies for control.

Citation: Deslyper G, Colgan TJ, Cooper AJR, Holland CV, Carolan JC (2016) A Proteomic Investigation of Hepatic Resistance to *Ascaris* in a Murine Model. PLoS Negl Trop Dis 10(8): e0004837. https://doi.org/10.1371/journal.pntd.0004837

Editor: Jason Mulvenna, Queensland Institute of Medical Research, AUSTRALIA

Received: February 3, 2016; Accepted: June 20, 2016; Published: August 4, 2016

Copyright: © 2016 Deslyper et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability: The MS raw data files and MaxQuant search output files have been deposited to the ProteomeXchange Consortium via the PRIDE partner repository with the dataset identifier PXD003555.

Funding: Funding for this work was obtained from MU Department of Biology and TCD Department of Zoology contributions to JCC and CVH respectively. A stipend to AJRC was obtained from the Welcome Trust Biomedical Vacation Scholarship and the Q-Exactive quantitative mass spectrometer was funded under the SFI Research Infrastructure Call 2012; Grant Number:

12/RI/2346 (3). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Introduction

Ascariasis is an important, widespread geohelminth disease of humans and pigs [1]. Over 800 million people are estimated to be infected with the causative agent, the human roundworm, *Ascaris lumbricoides* [2] and the equivalent in pigs, *Ascaris suum*, is equally ubiquitious [3]. The impact on children is particularly severe, with chronic morbidity such as growth retardation and diminished cognitive development being exhibited [4]. Transmission is linked to poor disposal of human waste leading to extensive contamination of the environment with long-lived, resistant eggs that embryonate under appropriate conditions of temperature and moisture [4].

The intensity of worm burden in the intestine of the host is a major determinant of the severity of disease [5]. Furthermore, the number of macroparasites a host carries is fundamental to our understanding of helminth parasite epidemiology [5]. Since the seminal work of Crofton [6], that characterized the frequency distribution of macroparasites as clumped or overdispersed, the ubiquity of what is now known as aggregation, has been demonstrated for a wide range of host-macroparasite systems [7]. Longitudinal field-based studies that measured the patterns of helminth re-infection in individual patients after the provision of anthelmintic treatment identified the phenomenon known as predisposition [8], whereby individuals demonstrated a degree of consistency in their patterns of re-infection and this was demonstrated from a range of endemic regions and geohelminth species [5]. Predisposition was also maintained over multiple rounds of chemotherapeutic treatment [9] and many years after a single round of treatment [10]. Aggregation and predisposition were also successfully modelled in outbred pigs infected with *A. suum* and were described as analogous to those of *A. lumbricoides* in humans [11].

The basis of this predisposition remains unknown [12] although it has been predicted that both exposure and host susceptibility are likely to influence the observed epidemiological patterns [5]. However, unravelling the relative contributions to aggregation and predisposition and hence susceptibility/resistance remains challenging for both ethical and logistical reasons [13]. As outlined by Keymer and Pagal [14], experimental manipulation utilising appropriate animal models is desirable, in tandem with human studies under field conditions, in order to study the multiple factors likely to be involved in predisposition.

Ascaris is a parasite that not only exists as an adult worm in the host intestine but also has a migratory pathway undertaken by its larvae, known as the hepato-trachaeal migration [15]. Symptoms occur during larval migration due to tissue damage [16] and the resultant pathology has been documented in the liver of both humans [17–19] and pigs [20–24]. Loeffler [25] described a transient or seasonal syndrome of pulmonary infiltrates, mild to marked respiratory symptoms and peripheral eosinophilia that he subsequently attributed to larval *Ascaris* in the lungs, termed Loeffler's syndrome [26]. Pulmonary symptoms can be severe and life-threatening [27] and have also been documented in pigs [28].

Despite its global prevalence and sheer numbers of individuals it infects, ascariasis remains a classic neglected disease [1] and part of the explanation for this neglect, is because pigs as animal models are costly and laborious and we lack the versatility of inbred strains. The majority of model organisms that have been infected with *Ascaris* are so-called abnormal hosts, whereby the parasite does not complete its life-cycle, but manifests itself as the early migratory phase [29]. Furthermore, in the vast majority of such experimental systems, the basis of resistance or susceptibility to infection has not been clearly established [29]. In contrast, a convenient and repeatable mouse model for exploring the susceptibility to *Ascaris* during the early phase of infection has been developed [30–32].

C57BL/6J are uniquely susceptible to infection with the porcine ascarid, *A. suum*, and larval burdens recovered from the lungs of this strain, markedly exceed those recovered from similarly infected strains of which the most resistant is the CBA/Ca mouse [30]. These two strains therefore represent opposite response phenotypes [5], mimicking the extremes of predisposition detected in humans and pigs.

Subsequently, this model was utilised to assess the significance of inflammatory processes within the murine lung. Such responses mirrored larval intensity and it was concluded that the pulmonary inflammatory immune response was not prominently involved in primary protection of mice to *Ascaris* infection in the lungs [31]. The lack of support for a pulmonary mechanism led to the suggestion that a hepatic/post-hepatic factor, which varies between C57BL/6J and CBA/Ca mice, may play a critical role in the successful migration through host tissues [31]. Evidence in the form of a differential histopathological change in the liver between the two strains was observed, whereby the resistant strain, CBA/Ca demonstrated an earlier intense inflammatory response coupled with a more rapid tissue repair in the liver [32].

C57BL/6 and CBA/Ca mice have also been used as model organisms to study susceptibility and resistance to other helminth and protist parasites. For example, in the case of *Brugia malayi*, the parasites were cleared more rapidly from the blood stream in CBA/Ca mice compared to C57BL/6J mice [33]. Studies involving *Leishmania*, demonstrated that CBA mice were able to control infection with *Leishmania major*, but not an infection with *Leishmania amazonensis* [34]. C57BL/6 mice conversely were found to be resistant to *L. major* infection, but were, as was the case for the CBA mice, susceptible to *L. amazonensis* [35]. *Schistosoma mansoni* caused low grade pathology in C57BL/6 mice, but a more aggravated pathology in their CBA counterparts [36]. In the case of *Plasmodium berghei* ANKA, both CBA and C57BL/6 were susceptible to the development of cerebral malaria [37]. In short, it is clear that both strains respond differently when challenged by a diversity of parasitic infections. However, depending on the parasite species, the mouse strains will exhibit a difference in resistance and susceptibility.

The aims of the present study were to investigate the differences in the liver proteomes between the two mouse strains, one deemed to be susceptible, the other resistant, with both control and infected groups within each strain. To address these aims we employed high throughput quantitative mass spectrometry (MS) which is routinely used to identify and quantify thousands of proteins from highly complex samples. We specifically used label-free quantitative (LFQ) mass spectrometry [38] on total protein extracts from the left liver lobes of C57BL/6J and CBA/Ca mice with and without *Ascaris* infection sampled on day 4 post-infection (p.i.). The hepatic lobe and time-point choice were based on the study of Dold *et al.* [32], who demonstrated relatively high and equivalent larval numbers in C57BL/6J and CBA/Ca left lobes at day 4 p.i., a time point at which they also observed the onset of a differential inflammatory response to infection between the two mouse strains. LFQ-based proteomics is increasingly being used to investigate human pathogens and parasites including *Chlamydia trachomatis* [39], *Acinetobacter baumannii* [40] and *Plasmodium vivax*. infected and uninfected erythrocytes [41] and is providing unprecedented insight into both specific and broad level interactions between the infectious agent and its host.

Materials and Methods

Ethics statement

The samples obtained for this study were part of a project authorised by the Health Products Regulatory Authority (HPRA), the competent authority that regulates scientific animal research in Ireland in accordance with Directive 2010/63/EU and its Irish transposition, SI No 543 of 2012 (Project Authorisation ID AE19136/P008 ID; Case Reference 7015826). In addition, this project was ethically approved by the TCD Animal Research Ethics Committee (AREC) prior to HPRA submission and approval.

Sample preparation

Both C57BL/6J and CBA/Ca mice (Harlan laboratories) were infected with 1000 embryonated ova of *A. suum* and euthanized using cervical dislocation on day four post infection (p.i.). At this time point, five animals were euthanised from all four groups: C57BL/6J infected, C57BL/6J control, CBA/Ca infected and CBA/Ca control [42]. The livers were extracted and each lobe was snap frozen separately using liquid nitrogen and stored at -80°C until required.

To confirm a differential larval burdens in susceptible and resistant strains, post-mortems of 5 infected mice from each strain were conducted on day 7 post-infection. Living *Ascaris* larvae were recovered from the lungs of each mouse by means of the modified Baermann technique [30]. A pellet of the isolated viable larvae was suspended in a 5 ml solution of 0.9% saline and 6% formalin. Prior to larval counts the 5 ml solutions were agitated to ensure a homogeneous distribution of larvae within the sample. Larval counts were recorded from lung samples by means of pipetting 2 mls of solution into the chamber of a nematode counting slide (Chalex corporation)[32]. The number of larvae in the gridded area, which represented 1 ml, was counted under X40 magnification. The number of larvae in a 1 ml solution was multiplied by the total volume in order to estimate the number of larvae in each lung sample.

The left lobes of day four p.i. were homogenized in 6M urea, 2M thiourea, supplemented with a protease inhibitor cocktail (Roche, Complete Mini). Samples were centrifuged for 5 min at 11,200 × g to pellet any cellular debris. The supernatant was then removed and quantified using the Qubit[™] protein quantification system (Invitrogen), following the manufacturer's instructions. Three independent biological replicates for each group were analysed in this study. 75 µg of each sample was precipitated using the 2D Clean-Up Kit (GE HealthCare), following the manufacturer's instructions. The resulting protein pellet was resuspended in 6M urea, 2M thiourea, 0.1 M Tris-HCl, pH 8.0. 50mM ammonium bicarbonate was added to each sample and proteins were reduced with 0.5M idthiothreitol (DTT) at 56°C for 20 min and alkylated with 0.55M iodoacetamide (IAA) at room temperature for 15 min, in the dark. 1 µl of a 1% w/v solution of Protease Max Surfactant Trypsin Enhancer (Promega) and 1 µg of Sequence Grade Trypsin (Promega) was added to give a protein:trypsin ratio of 75:1. The protein/trypsin mixture was incubated at 37°C for 18 hours. Digestion was terminated by adding 1 µl of 100% trifluoroacetic acid (Sigma Aldrich) and incubation at room temperature for 5 min. Samples were centrifuged for 10 min at 13,000 × g and a volume equivalent to 40 µg of pre-digested protein was removed and purified for mass spectrometry using C18 Spin Columns (Pierce), following the manufacturer's instructions. The eluted peptides were dried using a SpeedyVac concentrator (Thermo Scientific Savant DNA120) and resuspended in 2% v/v acetonitrile and 0.05% v/v trifluoroacetic acid for 5 min to aid peptide resuspension followed by centrifugation for 5 min at 16,000 × g. The supernatant was removed and used for mass spectrometry.

Mass spectrometry

1 µg of each digested sample was loaded onto a QExactive (ThermoFisher Scientific) high-resolution accurate mass spectrometer connected to a Dionex Ultimate 3000 (RSLCnano) chromatography system. The peptides were separated by a 2% to 40% gradient of acetonitrile on a Biobasic C18 Picofrit column (100mm length, 75mm ID), using a 120 minute reverse-phase gradient at a flow rate of 250nL min⁻¹. All data were acquired with the mass spectrometer operating in automatic data dependent switching mode. A full MS scan at 140,000 resolution and a range of 300–1700 m/z was followed by an MS/MS scan, resolution 17,500 and a range of 200–2000 m/z, selecting the 15 most intense ions prior to MS/MS.

Protein identification and LFQ normalisation of MS/MS data was performed using MaxQuant v1.5.0.8 (http://www.maxquant.org) following the general procedures and settings outlined in [43]. The Andromeda search algorithm [44] incorporated in the MaxQuant software was used to correlate MS/MS data against the SWISS-PROT database for *Mus musculus* [45](16,773 entries, downloaded May 2015) and a contaminant sequence set provided by MaxQuant. The following search parameters were used: first search peptide tolerance of 20 ppm, second search peptide tolerance 4.5 ppm with cysteine carbamidomethylation as a fixed modification and N-acetylation of protein and oxidation of methionine as variable modifications and a maximum of two missed cleavage sites allowed. False Discovery Rates (FDR) were set to 1% for both peptides and proteins and the FDR was estimated following searches against a target-decoy database. LFQ intensities were calculated using the MaxLFQ algorithm [46] from razor and unique peptides with a minimum ratio count of two peptides across samples. Peptides with minimum length of seven amino acids were considered for identification and proteins were only considered identified when more than one unique peptide for each protein was observed.

Data analysis

Perseus v.1.5.0.8 (www.maxquant.org/) was used for data analysis, processing and visualisation. Normalised LFQ intensity values were used as the quantitative measurement of protein abundance for subsequent analysis. The data matrix was first filtered for the removal of contaminants and peptides identified by site. LFQ intensity values were log₂ transformed [47] and each sample was assigned to its corresponding group (C57BL/6J control and infected; CBA/Ca control and infected). Proteins not found in two out of three replicates in at least one group were omitted from the analysis. A data-imputation step was conducted to replace missing values with values that simulate signals of low abundant proteins [48] chosen randomly from a distribution specified by a downshift of 2.6 times the mean standard deviation (SD) of all measured values and a width of 0.37 times this SD.

Two sample t-tests were performed for all relevant comparisons using a cut-off of p<0.05 on the post imputated dataset to identify statistically significant differentially abundant (SSDA) proteins. Volcano plots were generated in Perseus by plotting negative log p-values on the y-axis and log₂ fold-change values on the x-axis for each pair-wise comparison to visualise changes in protein expression. The 'categories' function in Perseus was utilized to highlight and visualise the distribution of various pathways and processes on selected volcano plots. Normalised intensity values were used for a principal component analysis (PCA). Exclusively expressed proteins (those that were uniquely expressed or completely absent in one group) were identified from the pre-imputation dataset and included in subsequent analyses. Hierarchical clustering was performed on Z-score normalised intensity values for all differentially abundant proteins by clustering both samples and proteins using Euclidean distance and complete linkage.

Gene ontology (GO) mapping was also performed in Perseus using the UniProt gene ID for all identified proteins to query the Perseus annotation file (downloaded January 2015) and extract terms for biological process, molecular function, Kyoto Encyclopaedia of Genes and Genomes (KEGG) name, KEGG pathway, protein family (pfam) and InterPro. GO and KEGG term enrichment analysis was performed on the major protein clusters identified by hierarchical clustering using a Fisher's exact test (a Benjamini-Hochberg corrected FDR of2%) for enrichment in Uniprot Keywords, gene ontology biological process (GOBP), gene

ontology cellular component (GOCC) and KEGG (FDR <2%). The Search Tool for the Retrieval of INteracting Genes/Proteins (STRING) [49] v10 (http://string-db.org/) was used to map known and predicted protein:protein interactions. UniProt gene lists (extracted from Perseus) were inputted and analysed in STRING using the medium to high confidence (0.5–0.7) setting to produce interactive protein networks for each group in all comparisons. GO term enrichment analyses for biological process, molecular function and cellular compartment were then conducted to identify potential pathways and processes that warranted further analysis. Such pathways were examined using the KEGG pathway analysis (http://www.kegg.jp/kegg/tool/map_pathway2.html) [50, 51], using the 'KEGG Mapper—Search&Color Pathway' tool. The equivalent KEGG identifiers were obtained using the UniProt 'Retrieve/ID mapping' function (http://www.uniprot.org/uploadlists/) with the organism set to *M. musculus* (mmu). Retrieved KEGG IDs were used to identify the most represented pathways. The MS proteomics data and MaxQuant search output files have been deposited to the ProteomeXchange Consortium [52] via the PRIDE partner repository with the dataset identifier PXD003555.

Results

The mean larval burden in the lungs of the C57BL/6J mice (n = 5) was 188 \pm SD 78.7 and the mean larval burden for CBA/Ca mice (n = 5) was 12 \pm SD 8.5 confirming the resistant and susceptible phenotype in the mice used in our experiment.

LFQ MS was performed on three replicates originating from the liver left lobes of C57BL/6J and CBA/Ca mice with and without *Ascaris* infection. 3,145 proteins were identified initially of which 2,307 remained after filtering and processing (S1 Dataset). Proteins, exclusive to or absent from an individual sample were included in subsequent analyses as proteins of significant interest (S1 Table). A principal component analysis (PCA) performed on all filtered proteins (Fig 1A) distinguished the C57BL/6J and CBA/Ca samples indicating a clear intrinsic difference between the two strains. Although the C57BL/6J infected and controls are well resolved there is some overlap within CBA/Ca infected and control groups, indicating that the response to *Ascaris* in C57BL/6J is perhaps more pronounced.



Fig 1. Label free quantitative proteomic analysis of liver left lobes of C57BL/6J and CBA/Ca mice with and without *Ascaris* infection. (A) Principal component analysis (PCA) of the left lobe liver proteomes of C57BL/6J and CBA/Ca mice with and *without* Ascaris infection. A clear distinction can be observed between the strains and the distinction between C57BL/6J infected and controls is greater than that for CBA/Ca infected and controls. (B) Two-way unsupervised hierarchical clustering of the median protein expression values of all statistically significant differentially abundant proteins. Hierarchical clustering (columns) resolved four distinct clusters comprising the three replicates from their original sample groups and five protein cluster (rows) based on expression profile similarities. (C) The table shows a selection of significantly enriched GOBP and GOCC terms (Fisher's exact test, FDR < 2%) in each of the identified protein clusters. https://doi.org/10.1371/journal.pntd.0004837.g001

Hierarchical clustering of z-score normalised intensity values for all significantly differentially abundant proteins (n = 194) resolved the three replicates of each sample group (Fig 1B). In addition, one minor and four major protein clusters were identified: the latter comprising C57BL/6J and CBA/Ca abundant proteins (Cluster A), C57BL/6J control and infected proteins (Cluster B); predominantly C57BL/6J and CBA/Ca infected abundant proteins (Cluster D) and CBA/Ca control and infected abundant proteins (Cluster E). GO and KEGG term enrichment analysis identified key biological processes enriched within each cluster (S2 Table; summarised in Fig 1C) and included translation and nucleosome (Cluster A); lysosome and glutathione metabolism (Cluster B); cytoskeleton, regulation of response to stress and extracellular vesicular exosome (Cluster D) and mitochondrial part and oxidation-reduction process (Cluster E).

Differential abundance and interaction network analysis

Performing two sample t-tests resulted in the identification of SSDA proteins both between and within strains (S2 Dataset). These differentially abundant proteins, together with proteins uniquely expressed in each group, were analysed using the interaction network analysis software, STRING, to identify biological pathways and processes over-represented in a particular group. Biological processes and pathways identified by STRING were investigated further for their representation within the differentially abundant protein dataset and displayed on volcano plots using the 'categories' function in Perseus to highlight proteins involved in selected biological processes.

C57BL/6J infected compared to C57BL/6J control

Two sample t-tests (p<0.05) on the post-imputation dataset identified 479 SSDA proteins between C57BL/6J infected and control with the log₂ fold change ranging from -8.2 to 7.2 (Table 1). The 20 most differentially abundant proteins for both groups are displayed in Fig 2A. The most abundant proteins in C57BL/6J controls were histone H1.4, tyrosine aminotransferase, and histone H1.1. In comparison, for C57BL/6J infected samples, the most abundant proteins consisted of two S100 proteins (protein S100-A8) and protein S100-A9) and cytochrome c oxidase subunit 7C protein. STRING analysis identified enrichment of terms associated with the proteasome, the mitochondrion, RNA splicing and actin cytoskeleton in C57BL/6J infected individuals (Fig 3A) whereas enriched terms were identified for translation in the C57BL/6J controls (Fig 3B).

	Number of proteins*	Range of FC Difference	Protein Processes					
T-text Comparison			Milochondria (324)	OXPHOS (NI)	Immune system (198)	Ribosomes (73)	tron lon binding (71)	
CS78L/6J Mixtel CS78L/6J control	456479	4272		22	20	48	21	
CBA/Caintected CBA/Ca control	108793	6153	.17	8	14	35	- 5	
CBA/Calcontext CS25L/AU control	320/364	6.7113.6	72	25	14			
CBA/Ca infected: C578L/ 6J infected	376410	4162	78	20	15	18	14	
· pre-imputation/post imputa	ion							
to: 10 1071 Bournal antis 5004857 S	101							

Table 1. Numbers of statistically significant differentially abundant (SSDA) proteins (t-tests; p<0.05) identified among CBA/Ca and C57BL/6J infected and uninfected samples.

The number of proteins before and after imputation, the range of Log₂ fold differences and the number of proteins associated with well represented KEGG or gene ontology processed or pathways are given. Numbers in parenthesis represent the total number of proteins identified within each process. OXPHOS: oxidative phosphorylation, FC: fold change. https://doi.org/10.1371/journal.pntd.0004837.t001



Fig 2. Volcano plots of all identified proteins based on relative abundance differences between C57BL/6J and CBA/Ca mice with and without *Ascaris*.

Volcano plots showing the distribution of quantified proteins according to p value ($-\log_{10} p$ -value) and fold change ($\log_2 mean LFQ$ intensity difference). Proteins above the line are considered statistically significant (p-value <0.05) and those to the right and left of the vertical lines indicate relative fold changes ≥ 1.5 . The top 20 differentially abundant proteins are annotated and all proteins associated with the mitochondria (blue), immune system (green) and translation (orange) are highlighted for (A) C57BL/6J infected and control, (B) CBA/Ca infected and control mice, (C) CBA/Ca control and C57BL/6J control and (D) CBA/Ca infected mice.

https://doi.org/10.1371/journal.pntd.0004837.g002



Fig 3. Interaction network analysis of up regulated proteins in C57BL/6J and CBA/Ca mice with and without Ascaris infection.

Protein interaction information was obtained from the STRING database using gene lists extracted for statistically significant differentially abundant (SSDA) proteins from pair wise t-tests (p< 0.05). Each node represents a protein and each connecting line represents an interaction, the extent of evidence for which is represented by the width of the line. Statistically enriched KEGG and biological process Gene Ontology (GO) descriptors were examined to identify clusters of proteins enriched within SSDA protein of increased abundance for (A) C57BL/6J infected in comparison to C57BL/6J controls, (B) C57BL/6J controls in comparison to C57BL/6J infected, (C) CBA/Ca infected in comparison to CBA/Ca controls and (D) CBA/Ca controls in comparison to CBA/Ca infected mice.

https://doi.org/10.1371/journal.pntd.0004837.g003

CBA/Ca infected compared to CBA/Ca control

Two sample t-tests on the post-imputation dataset identified 193 SSDA proteins (log₂ fold change range: -6.1 to 5.9; Table 1) with the 20 most differentially abundant proteins displayed in Fig 2B. The most abundant proteins in CBA/Ca infected mice were: protein S100-A8, seminal vesicle secretory protein 4, and dipeptidyl peptidase 1. For CBA/Ca control mice the most SSDA proteins included: 7-alpha-hydroxycholest-4-en-3-one 12-alpha-hydroxylase, costars family protein ABRACL, and collagen alpha-1(I) chain. STRING analysis revealed that CBA/Ca infected mice had an overrepresentation of spliceosome and actin cytoskeleton proteins were enriched in addition to RNA processing and glutathione metabolism in CBA/Ca control mice with respect to their infected counterparts (Fig 3D).

CBA/Ca control compared to C57BL/6J control

Two sample t-tests (p<0.05) on the post-imputation dataset identified 354 SSDA proteins respectively and the Log₂ fold changes ranged from -6.7 to 13.6 (Table 1). The top 20 most SSDA proteins are shown in Fig 2C. The most abundantly expressed proteins observed in CBA/Ca control samples are haemoglobin subunit beta-2, S-methylmethionine, and putative hydroxypyruvate isomerase. The log₂ fold change for haemoglobin subunit beta-2 was the highest observed in the dataset, being 13.6. For C57BL/6J controls, the most abundantly expressed proteins are aldehyde oxidase 3, ADP-ribosylation factor 5, and H-2 class I histocompatibility antigen K-B alpha chain. Two clusters were observed in the CBA/Ca control samples representing ribosomal and oxidative phosphorylation biological processes (S1A Fig). However STRING analysis failed to identify any well populated clusters in C57BL/6J compared to CBA/Ca control (S1B Fig).

CBA/Ca infected compared to C57BL/6J infected

Two sample t-tests (p<0.05) on the post-imputation dataset identified 410 SSDA proteins, respectively and the log₂ fold changes ranged from -8.1 to 9.2 (Table 1). The 20 most SSDA proteins are displayed in Fig 2D. The most abundant proteins in CBA/Ca infected compared to C57BL/6J were found to be: putative hydroxypyruvate isomerase, S-methylmethionine, haemoglobin subunit beta-2. The most abundant proteins present in C57BL/6J infected compared to their CBA/Ca counter parts were: murinoglobulin-2, aldehyde oxidase 3, and chitinase-3-like protein 3. STRING analysis resolved clusters with enriched terms for oxidative phosphorylation and translation in CBA/Ca infected mice (S1C Fig). Terms associated with the glutathione metabolism are enriched in C57BL/6J infected mice with respect to CBA/Ca infected mice (S1D Fig).

Enriched processes and pathways

Proteins involved in the processes and pathways of interest (identified in Perseus and STRING) were displayed on volcano plots arising from the 2-way t-tests to give an indication of expression profile for all significant and non-significant representatives. The five processes that were consistently overrepresented and/or differentially abundant and selected for further analysis were mitochondria, electron transfer chain, immune system, ribosome and iron ion binding (Table 1). KEGG analysis was also used to provide a protein centric view for selected pathways (S2–S4 Figs).

Mitochondria and associated processes

Mitochondrial associated proteins were consistently differentially abundant among all samples. Of the 2,307 proteins identified and reported here, 324 were associated with the mitochondrion. A higher abundance of mitochondrial proteins was generally observed in CBA/Ca mice both with and without infection (Table 1; Fig 1C) in comparison to C57BL/6J mice, although an increase in mitochondrial protein abundances was observed for both strains under infection (Fig 2). GO analysis in Perseus resolved 80 oxidative phosphorylation (OXPHOS) terms, including electron transfer chain (ETC) complex I to V proteins, the most differentially abundant of which were cytochrome c oxidases (cox7c and cox7a2). CBA/Ca control and infected mice had a higher abundance of proteins involved in the OXPHOS pathway, when compared to C57BL/6J (S2A Fig). Although there was a significant upregulation of OXPHOS proteins in C57BL/6J infected mice when compared to their controls, the extent of increase was less than that for CBA/Ca. KEGG analysis also indicated that there was a generally lower abundance of ETC proteins in C57BL/6J infected mice, compared to their CBA/Ca counterparts. However Complex I proteins, were less abundant in C57BL/6J infected mice, when compared to their controls (S2A Fig).

Proteins of the tricarboxylic acid (TCA) cycle were intrinsically more abundant in CBA/Ca control mice, when compared to their C57BL/6J counterparts. This difference between the two strains was observed also under infection. When infected, both mouse strains had increased abundance of these proteins. However, C57BL/6J infected mice had higher abundances for the entire TCA cycle (S3 Fig).

Translation

A clear change in the abundance of proteins involved in the translational process during infection was observed. Ribosomal proteins were more abundant in both CBA/Ca controls and infected with respect to their C57BL/6J counterparts (S1A and S1C Fig). Under infection however, both CBA/Ca and C57BL/6J mice had significantly lower abundances for many ribosomal proteins (Fig 2). KEGG analysis further confirmed this finding demonstrating that CBA/Ca had more intrinsically abundant ribosomal proteins than C57BL/6J, a difference that became more pronounced under infection (S2B Fig).

Immune system and response to stress

GO and KEGG term enrichment analysis indicated that there were very few immune associated pathways respondent to infection. Of these, 20 immune associated proteins were differentially abundant (relative fold difference > 1.5) between infected and control samples (Table 2). Eight proteins: s100-A8; s100-A9; dipeptidyl peptidase 1; coronin-1A; galectin; vitronectin; moesin and plasminogen were more abundantly expressed in both C57BL/6J and CBA/Ca infected mice with respect to their uninfected counterparts indicating a potential conservation of response to *Ascaris* in both mice strains. The majority of SSDA immune proteins were observed in C57BL/6J *Ascaris*-infected mice (Table 2). A number of immune proteins demonstrated higher intrinsic abundances in C57BL/6J compared to CBA/Ca (in both infected and uninfected comparisons) and included dipeptidyl peptidase 1, complement C4-B, galectin-3 and the MHC class I molecules H2-D1 and H2-K1 (Fig 2C and 2D).

Gene	Protein names	Unipeut ID	T feel Comparison	Legit fuld Difference	Inmune Function	
5100a8	Protein S100-AB	927006	C57/CBA	72.59	Timale Immune response, Laukoopte nigration in Inflammatory response, Neutrophil chemotexie, Positive regulation of inflammatory response, Autophagy: Positive regulation of aprentitie signaling	
\$100#9	Protein \$100-A8	P31725	CS7; DBA	4.1;47	Actin cytoskeleton reorganisation: Laukocyte chemotrasic. Positive regulation of inflammaticny response: Positive regulation of blood coagulation; Autophagy	
Obc	Dpoptidy poptideos 1	P97821	CBA CS7	55.13	T cell mediated cylclosicity, Positive regulation of apoptotic signaling pathway	
C pero	Galactin-3	P18110	OBA, C57	55.17	by binding Econophilinacrophage/monocytemeutrophil Cherostaxis: Invate invisure response	
Anual	Annexin A1	P92507	CI17	4.4	Granubognemonocyte chemotaxie, Phagocytosis, Negalive regulation of Tinapor 2 out differentiation, Navdrephil homoscalassi, Pankhvi regulation of Tinapor 1 cell differentiation, Positive regulation of neutrophil populotic possese, regulatoria Mitmentiation.	
Contra	Coronie-1A	089013	C8A, C57	37,30	Leukocyte chemotaxis. Natural killer cell degranulation Phagocytosis. Cellular response to insideukin-4. Positive regulation of T cell activation and proliferation	
Lord	Neutrophil gelatinase- associated toocalm	P11072	CI57	3.4	Cellular response to hydrogen percetde. Cellular response to interlector 1. Incute interlector response.	
in .	Vitrumentin	P29788	OS7 CBA	2.01.7	Immune response. Positive regulation of smooth muscle cell migration. Positive regulation of receptor metalated	
103	Interlaukin enhancer-binding factor 3	QSZ1X4	CBA	2.0	Defense response to virus	
Pp.1	Partition heaving chant	P00529	C17	1.1	Instrume response. Hogalive regulation of cell proliferation. Negative regulation of necrotic cell meth-	
Arap	Alpha 2 HS-glycoprotein	P29600	CB7	0.8	Acute-phase response, Positive regulation of phagocytosis, Regulation of inflammatory response	
inge 4	Interleukin-1 receptor- executed timese 4	085442	CSP.	0.9	Neutroph mediated immunity, JNK cascade. Positive regulate of Learned known/M Aspeals streating	
Lawlet	Galectin-1	P18045	CSP	0.8	Positive regulation of Happed kinase/NF-kepped signaling T cell costimulation	
Mari	Moesin	P25041	EST: CBA	08.04	Leukocyte cell-cell adhesion: Leukocyte migration	
Read?	Anvesin A2	P07366	CB7	0.8	Angegenesia, Ethnodysis, Positive regulation of Nonbiast profileration	
Pag .	Plasningen	P20918	CS7 CBA	0.62.8	Mononuliear cell register, Negative regulation of angiogenesis. Tesue regeneration	
Palanton	Passed activiting tacky acaty/hydrolase ID subunit alpha	P63005	410	0.2	Actin cytoskaletice organization, Negative regulation of JNK cascade. Positive regulation of cytokine-mediated signaling Regulation of microtubule motor activity	
Segentz	Alpha-2-ontplaamin	Q61242	CIS7	0.8	Acute-phase response. Positive regulation of stress fibre assembly: Regulation of blood vessel size by revin anglotemen heightive regulation of plasminogen activation	
rex.	Hengesin	(281)(72	CE1	0.6	Nerve metabolic process, Positive regulation of immunoglobulin production, Positive regulation of insuferon gamma mediated signaling, Positive regulation of response to interferon	
CEa	Complement component CB alpha chain	09082	C57; DBA	04.05	Complement activation; Cytolysis	

Table 2. Statistically significant differentially abundant (SSDA) proteins associated with immune processes.

Log₂ fold differences from two pairwise t-tests (p < 0.05) are given for (i) CBA: CBA/Ca infected and CBA/CaC57 control and (ii) C57: C57BL/6J infected and C57BL/6J control. The number of immune associated proteins and relative fold change differences are for the majority greater in C57BL/6J than CBA/Ca *Ascaris*-infected mice. Immune functions were retrieved by searching the UniProt ID on the UniProt Knowledgebase (http://www.uniprot.org/). https://doi.org/10.1371/journal.pntd.0004837.t002

16 proteins associated with the biological process term "response to stress" were enriched in the shared "response to infection" cluster (Cluster D on Fig 1B). Many of these proteins are also associated with the cytoskeleton and cytoskeletal organisation, terms that were also shown to be significantly enriched within this "infected" cluster (S2 Table; Fig 1C).

Role of glutathione metabolism in resistance

Glutathione is known to reduce the reactive oxygen species (ROS), H_2O_2 to H_2O . Because of its ROS reducing capacities, it was investigated in this study. KEGG analysis identified an increase in glutathione metabolic proteins of C57BL/6J control samples, compared to their CBA/Ca counterparts (S4 Fig). This inherent difference was also observed when comparing both strains under infection. Additionally, there was an increase in abundance of glutathione metabolism proteins in C57BL/6J mice when infected, compared to their controls. This increase was less pronounced in CBA/Ca mice.

Discussion

With 800 million people worldwide infected with *Ascaris* [2], improving our understanding of the underlying biological pathways involved with susceptibility to infection is required. Currently, ascariasis control consists primarily of the provision of anthelmintic drugs, such as albendazole or mebendazole [53], however, reinfection can occur rapidly after treatment [54]. Furthermore, the presence of adult worms in the intestine contributes to significant chronic morbidity and, in some cases, more acute complications [4]. Therefore, investigation of the role of the liver—a key organ in *Ascaris* larval migration, and likely attrition and consequent establishment of adult worms in the intestine could prove fruitful [29]. The liver has previously been identified as the organ of interest to investigate susceptibility and resistance to *Ascaris* infection [31, 32]. To understand the difference in susceptibility and resistance to heavy infection, we have performed a proteomic analysis of the liver of two different mouse strains, C57BL/6J and CBA/Ca, models for susceptibility (heavy larval burden) and resistance (light larval burden), respectively. The difference in susceptibility was confirmed by assessing larval burdens in the lungs C57BL/6J and CBA/Ca mice on day 7 p.i. Previous studies have demonstrated a distinct, repeatable difference in *A. suum* larval burden in the lungs of susceptible C57BL/6J and resistant CBA/Ca inbred mice on day 7 post-infection [30, 32, 55]. Although the numbers observed were very similar to those described in these studies, a clearer divergence in larval numbers was observed between both strains in the present experiment.

Using quantitative mass spectrometry, several pathways and individual proteins, which may contribute to these observed differences in the burden of infection, were identified. A clear difference in the proteomes of both strains was observed (Fig 1A) and intrinsic differences identified between the inbred strains, unrelated to *Ascaris* infection, which may underpin susceptibility and resistance, and are also significant for other branches of biomedical research using these particular mice strains.

Mitochondrial differences potentially determine resistance or susceptibility

Parasite infection can place additional stresses on the physiological processes of the host. One of the major findings of this study was the higher abundance of proteins involved in mitochondrial processes in the CBA/Ca mouse strain, both with and without infection. These processes include OXPHOS and the TCA cycle, which were significantly enriched in the CBA/Ca proteome in comparison to their C57BL/6J counterparts. These intrinsic differences may represent a significant difference between both strains in terms of the susceptibility to *Ascaris* infection. The mitochondria are the main source of ROS, which are produced as by-products in the OXPHOS pathway as part of cellular respiration [56, 57]. ROS have a vast range of biological functions and play a role in defence against parasites and pathogens, apoptosis, cell survival, cell growth, proliferation, differentiation and many other signalling pathways [57, 58]. Higher ROS production in CBA/Ca mice may generate a prohibitive oxidative environment for *Ascaris* within the murine liver and explain the decrease in observable migration to the lungs by *Ascaris* from the CBA/Ca strain [32]. Our results support previous findings that CBA/Ca mice have higher superoxide production and have a higher tolerance to ROS than their C57BL/6 counterparts [59, 60]. Staecker *et al.* [60] found that the mouse cochlea of the C57BL/6J strain showed an early-onset hearing loss pathology compared to CBA/Ca mice, potentially linked to increased levels of free radicals present in the cochlea. This increase was attributed to the *adult hearing loss (Ahl*) gene, first identified by Johnson *et al.* [61], which is thought to cause a decrease in protective antioxidant enzymes in C57BL/6J mice. Taken together, these studies suggest that CBA/Ca mice not only produce higher endogenous levels of ROS and protective antioxidants but can tolerate ROS levels that could potentially affect an invading parasite.

Under infection, both strains exhibited higher abundances of OXPHOS and TCA cycle proteins indicating a general stress response to *Ascaris* infection. Although the increase in abundance was generally more pronounced in C57BL/6J mice, when compared to

their corresponding controls, infected CBA/Ca mice had higher overall abundances representing the higher endogenous levels of mitochondrial proteins in CBA/Ca control mice. Interestingly, C57BL/6J infected mice had lower abundances of complex I proteins specifically in comparison to their controls. Complex I is an important site of superoxide production (together with complex III) [62] and has been linked with a number of diseases spanning early childhood (Leigh disease) [63] to adulthood (Parkinson's and Alzheimer's disease) [62]. Additionally, quinone-binding inhibitors, which inhibit complex I, were shown to increase ROS production [62]. TCA cycle proteins showed a similar pattern of abundance changes amongst samples, with higher intrinsic abundances observed in CBA/Ca mice and increased abundances observed under infection. However, C57BL/6J infected mice displayed higher abundances of TCA-associated proteins under infection than CBA/Ca infected mice, compared to their controls. The TCA cycle takes place in the mitochondria [64] and results in the production of the reduced co-enzyme nicotinamide adenine dinucleotide hydride (NADH), which contributes electrons to the OXPHOS pathway. Thus, it seems that in response to infection both mice strains increase OXPHOS activity.

The differences in mitochondrial protein abundance and presumed ROS levels between the susceptible and resistant strains can be explained not only by intrinsic basal differences but also the extent of parasitic interaction within each strain. Based on the significant reduction in nematode numbers that reach the lungs in CBA/Ca mice [31] one can assume that the CBA/Ca-Ascaris interaction represents a far less compatible host-parasite interaction in comparison to the C57BL/6J-Ascaris interaction. Nematodes in C57BL/6J livers seem capable of successful establishment within their hosts and in doing so, modulate host defence pathways and responses to allow for completion of the parasite life-cycle. *Ascaris* secrete a plethora of proteins in their excretory/secretory fluid with potential immunomodulatory and antioxidant defence functions [65–67]. The first complete predicted secretome was generated during the *Ascaris suum* genome project identifying 775 predicted secretory proteins with a rich number of peptidases for the migration through host tissues, as well as products potentially involved in the evasion or modulation of host defences [68]. In addition, proteomic analyses of the excretory/secretory products of larval and adult *Ascaris* have characterised heat shock proteins, ABA-1, proteases, serpins, chitinases and a suite of individual proteins that have been implicated in immune evasion and modulation and have been suggested to play a role in parasite survival [69]. It is not unreasonable to postulate that *Ascaris* also secrete proteins that reduce host mitochondrial processes and ROS production specifically. Modulation of host ROS production has been widely reported for a number of taxonomically diverse parasites, including the liver fluke, *Fasciola hepatica* and the causative agent of Chagas disease, *Trypanosoma cruzi* [70–73]. In addition, antioxidant production in parasites has been linked to improved parasite survival [74, 75] and helminths are known to possess an array of ROS-reducing products, suc

An intrinsic difference between the two mouse strains was also observed for glutathione metabolism, with C57BL/6J mice displaying a higher abundance in proteins involved in this pathway. Under infection, both strains displayed an increase in abundance for these proteins, with C57BL/6J mice presenting a more pronounced upregulation. Most mitochondria lack CAT, therefore, H₂O₂ conversion is maintained by glutathione (GSH) within these organelles [79]. H₂O₂ is reduced to H₂O by glutathione peroxidase (GPx), using GSH [80] and this reaction produces oxidised glutathione (GSSG), which is then reduced back to GSH by glutathione reductase (GR), consuming one molecule of NADPH. This increase would need to be countered by increased antioxidants, such as GSH.

Our findings clearly indicate that mitochondrial- and ROS-associated proteins are more abundantly expressed in both strains under infection, indicative of the host liver being within a state of stress. In C57BL/6J mice, these increases are less pronounced than in CBA/Ca mice indicating that although a response to *Ascaris* is mounted, the levels achieved in C57BL/6Js are not equivalent to those in CBA/Ca mice. In addition, because ROS levels are presumably reduced in C57BL/6J mice, the nematodes entering the liver have a less cytotoxic environment to contend with and can potentially mount immuno-modulatory and antioxidative perturbations on their host.

A more pronounced immune response is observed in C57BL/6J mice

The host immune system represents an important obstacle to the completion of the parasite life-cycle. To circumvent this threat, parasites have evolved strategies to evade or modulate aspects of the host immune response. In relation to *Ascaris*, previous studies examining the host immune response within the liver have focussed on the role of pro-inflammatory cytokines in co-ordinating defences against the parasite during its migratory phase [31, 81, 82]. Within the present study, gene ontology and KEGG pathway analyses failed to identify any clear process level immune responses to *Ascaris* infection and few differentially abundant immune proteins were identified either between strains or in response to the presence of *Ascaris*. The low number of immune-responsive proteins may be explained in two ways. Firstly, the liver is an immune tolerant organ and the hepatic system circumvents the lymphatic system. This ensures that no unwanted immune activation occurs in response to commensal pathogens present in the bowel, which can sometimes enter the blood stream after endothelial damage [83]. This system is thought to be exploited by other parasites, such as malaria-causing *Plasmodium* species, where the parasite enters the liver in the present study may be too early in the infection to observe an adaptive immune response, as the adaptive immune response takes 4–7 days to engage fully [85]. Of the identified differentially abundant immune proteins, broadly three groups can be defined (see Table 2): proteins observed in both mouse strains, protein abundance only observed in C57BL/6J mice, and proteins observed in C57BL/6J mice, and proteins observed in ceptone in response to infection, which was to be expected, and C57BL/6J mice displayed a higher abundance in some immune-associated proteins.

The proteins observed to be more abundant in both strains signify a potential innate immune response through the identification of proteins, such as complement component C8a and S100 proteins, S100a8 and S100a9. The alpha polypeptide of complement 8 interacts with other complement proteins to form the membrane attack complex, a cell-killing structure [86]. The complex binds to the cell membrane of target cells, forming transmembrane channels resulting in cell lysis and subsequent death [86]. Although, complement recruitment can occur in response to tissue damage potentially associated with parasite infection, complement components have previously been implicated as factors in mediating the adherence of myeloid cells to nematode parasites, resulting in parasite death although the susceptibility amongst different nematode species is variable [87]. Secretory products released by nematodes, such as *Brugia malayi* and *Trichinella spiralis*, have been identified to inhibit the chemotaxis activities of complement component 5a, identifying the complement component as a target for immunomodulation by parasitic nematodes [88].

Proteins S100a8 and S100a9 were among the most abundantly expressed under infection in both mouse strains, and completely absent in the control samples (Fig 1C and 1D). The S100 group is involved in multiple cellular functions, including cellular contraction, motility, cell differentiation and calcium regulation [89]. S100a8 and S100a9 can exist as homodimers, but in the presence of calcium, they will form a heterodimer called calprotectin [90, 91] which is expressed mainly by granulocytes, monocytes, and macrophages [90]. These proteins were recently identified as danger-associated molecular patterns (DAMPs) [92] which have been shown to interact with toll-like receptors (TLRs), such as TLR4 [93]. It is thought that S100a8/S100a9 are actively released by cells sensing danger, rather than passively [90]. Calprotectin has been found to be a chemotactic factor for neutrophils and other mononuclear cells, with the exception of lymphocytes, in mice [94], further confirming its role in the immune system.

S100a8 and S100a9 heterodimers are known to be important in *Leishmania* infection. S100a8 and S100a9 recruitment results in the elimination of *Leishmania* and knockout mice experiencing a more severe infection. Such proteins probably have a role in the recruitment of neutrophils to the site of infection [91]. Moreover, S100a8 and S100a9 primed macrophages were better at killing *Leishmania* than non-primed macrophages. In addition, Edgeworth *et al.* [95], using human biopsies, found that S100a8/S100a9 heterodimer secretion, by macrophages, onto adults of the filarial nematode, *Onchocerca volvulus*, was part of an early immune response—demonstrating a putative immune role against nematode parasites. A similarly early response may be signified by the presence of these proteins in infected C57BL/6J and CBA/Ca mice suggesting a conserved role in defence against parasitic nematodes. In short, our results suggest the presence of an innate immune response (see Table 2) through the presence of a complement protein (C8a) and several proteins involved in leukocyte chemotaxis (such as S100a8, S100a9, galactin-3). Additionally, there is evidence of an early adaptive immune response through the presence of proteins such as plastin-2 (Lcp1) and dipeptidyl peptidase 1 (Ctsc).

Given that C57BL/6J mice carry more active *Ascaris* than CBA/Ca mice, it was unsurprising to find that a higher number of immune-associated proteins were observed in the susceptible strain. Of the nineteen proteins with known immune function (Table 2) fifteen had higher fold changes in C57BL/6J compared to CBA/Ca mice or were exclusively more statistically abundant in C57BL/6J mice. The two most abundantly expressed proteins in the 'C57BL/6J' only group, were annexin A1 and neutrophil gelatinase-associated lipocalin, both of which are expressed as a cellular response to interleukin-1 (IL-1). It seems therefore that the restrictions encountered by the nematode larva in the resistant strain results in the reduced immune response in comparison to their susceptible counterparts.

Reduction in the translation machinery occurs under infection

Very little is known about how extracellular macroparasites, such as *Ascaris*, successfully modulate their host. The results of this study suggest that under infection, a down regulation of translational proteins occurs for both strains, in particular S6 ribosomal protein (part of the mammalian/mechanistic target of rapamycin (mTOR) pathway). Additionally, CBA/Ca control mice had, both with and without infection, higher ribosomal protein abundances compared to C57BL/6J mice. Under infection this intrinsic difference, however, became more pronounced. Pathogens are known to interact with translational proteins in their host [96]. Viruses, for example, are dependent on the host translational machinery for their own protein synthesis. Cells in turn can promote gene expression in response to the environmental situation, e.g. hypoxia, glucose deprivation [96] and pathogens, such as certain bacteria, are known to secrete effectors into the cytoplasm which in turn can reduce translation itself [97]. Inactivation of translation machinery is a documented strategy of certain parasites, such as the trypansome, *Leishmania major* [98]. Once the parasite has invaded a macrophage, the surface protease GP63 cleaves mTOR and mTOR complex 1 (mTORC1), which inhibits translational initiation [98]. It is reasonable to postulate that the widespread lower abundance of translation proteins may indicate direct modulation by *Ascaris* which is known to secrete a wide range of effector-like molecules [69]. However it must be acknowledged that the down regulation of translation may in fact signify a defensive strategy employed by the host, in response to damage caused by *Ascaris*, as is seen in bacteria-induced epithelial damage, which results in the triggering of signals that suppress host translation [96].

Haemoglobin profiles distinguish both strains

Aged erythrocytes are broken down by macrophages in the spleen and liver [99]; the liver is therefore an important organ in haemoglobin metabolism. Haemoglobin beta-2 (Hbb2) is the most abundantly expressed protein in CBA/Ca mice, and displayed relative fold changes of over 12,500 between CBA/CA and C57BL/6J controls and a fold chance of over 500 between CBA/CA and C57BL/6J infected mice. The haemoglobin of mice consists of two haemoglobin alpha (Hbba), one haemoglobin beta-1 (Hbb-1, major chain) and one Hbb-2 (minor chain) subunits. There are three different haplotypes for haemoglobin beta (Hbb): HbbS (single, β^S), HbbD (diffuse, β^D) and HbbP. C57BL/6J mice are homozygous for HbbS [100], whereas CBA mice are hetrozygous for HbbD [100]. The β^S haplotype has one reactive cysteine residue on position 93 (βCys93), whereas β^D and β^P have an extra reactive cysteine residue: βCys13 [100]. Interestingly, both βCys93 and βCys13 can be modified by GSSG, with βCys13 being more susceptible than βCys93 [100]. Furthermore, GSSG may be reduced to GSH at βCys13, without the need of GR nor NADPH [100]. Hempe *et al.* [100] postulate that the concentration of these cysteine sulfhydryl groups determines the availability of GSH for enzymatic reactions. Having a higher concentration of haemoglobin could therefore be a mechanism of CBA/Ca mice to establish their ROS tolerance (as observed in the previously mentioned radiation studies), as a higher haemoglobin concentration would coincide with more β Cys93, and thus more chances for glutathione to be reduced. The allelic differences in haemoglobin are also thought to confer a differential ability of various mouse strains to contend with different parasites. For example, C57 mice (HbbS) are relatively resistant to *Plasmodium* infection compared to BALB/c mice (HbbD)[103].

Conclusion

Ascariasis is a debilitating disease affecting an estimated 800 million individuals globally. While the pathology of the disease has been extensively examined, our understanding of the molecular mechanisms underlying resistance and susceptibility to nematode infection is poorly understood. Here we provide a novel insight into the changes in a host liver proteome in response to Ascaris infection *in vivo* within two murine strains varying in their resistance and susceptibility to infection. Our results provide evidence for significant intrinsic differences in the hepatic proteomes of both mouse strains, potentially associated with resistance to Ascaris infection. Given the higher levels of proteins associated with ROS producing and processing and the general increased tolerance to ROS in CBA/Ca mice [59] in comparison to C57BL/6J mice, in particular, we hypothesise that higher ROS levels and the associated oxidative environment could be involved in the inhibition of Ascaris larval in CBA/Ca mice. Whether the intrinsic differences in mitochondrial protein abundances are due to different levels of mitochondrial biogenesis and number between the two strains has yet to be determined. Our research provides new insights into the intricacies and complexities of the host-parasite relationship of Ascaris. In addition, potential parasite modulation of translational processes by Ascaris were clearly evident in both strains. Our findings also provide a new understanding of previous studies that utilised these two mouse strains for experiments involving early-onset hearing loss, radiation exposure, and several other micro and macroparasite infections. Given our findings and the central role of the liver in the Ascaris migratory pathway, we suggest a potentially novel research direction to develop alternative preventative control strategies for Ascaris. It seems that the key determinant in murine resistance to Ascaris lies in highly oxidative conditions that presumably restricts and arrests successful larval migration within the CBA/Ca hepatic environment. Larval nematodes that enter the C57BL/6J liver seem free to continue their onward progression and through the secretion of their excretory/secretory compounds further sustain their parasitic lifecycle through manipulation and modulation of the host liver. So although defence responses are mounted in C57BL/6J mice it seems that Ascaris is already well-established in its attempts to contend with the host response. However, through the manipulation of hepatic ROS levels in the susceptible mouse strain, we may now be able to determine the importance of intrinsic ROS in conferring resistance to Ascaris. Although significant research is required to fully understand the determinants of resistance to Ascaris in our murine model, it does seem that we have at least been presented with new options in our pursuit of strategies to control a disease that affects an estimated one eighth of our planet's population.

UniProt accession numbers for proteins mentioned in text

7-alpha-hydroxycholest-4-en-3-one 12-alpha-hydroxylase: O88962; ADP-ribosylation factor 5: P84084; Aldehyde oxidase 3: G3X982; Annexin A1: P10107; Chitinase-3-like protein 3: O35744; Collagen alpha-1(I) chain: P11087; Complement C4-B: P01029; Complement component C8 alpha chain: Q8K182; Coronin-1A: O89053; Costars family protein ABRACL: Q4KML4; Cytochrome c oxidase subunit 7A2, mitochondrial: P48771; Cytochrome c oxidase subunit 7C, mitochondrial: P17665; Dipeptidyl peptidase 1: P97821; Galectin-3: P16110; H-2 class I histocompatibility antigen K-B alpha chain: P01901; H-2 class I histocompatibility antigen, K-B alpha chain: P01901; H-2 class I histocompatibility antigen, M-B alpha chain: P01901; H-2 class I histocompatibility antigen, K-B alpha chain: P01901; Haemoglobin alpha: P01942; Haemoglobin beta-1: P02088; Haemoglobin subunit beta-2: P02089; Histone H1.1: P43275; Histone H1.4: P43274; Interleukin-1 beta: P10749; Mitochondrial antiviral-signalling: Q8VCF0; Moesin: P26041; Murinoglobulin-2: P28666; Neutrophil gelatinase-associated lipocalin: P11672; Plasminogen: P20918; Plastin-2: Q61233; Protein S100-A8: P27005; Protein S100-A9: P31725; Putative hydroxypyruvate isomerase: Q8R1F5; S6 ribosomal protein: P62754; Seminal vesicle secretory protein 4: P18419; S-methylmethionine: Q91WS4; Tyrosine aminotransferase: Q8QZR1; Vitronectin: P29788. UniProt accession numbers are provided for all identified proteins in S1 and S2 Datasets.

Supporting Information

S1 Dataset. Proteins identified from the liver left lobes of C57BL/6J and CBA/Ca mice with and without Ascaris infection.

https://doi.org/10.1371/journal.pntd.0004837.s001 (XLSX)

S2 Dataset. Statistically significantly differentially abundant proteins (2 sample t-tests; p<0.05) and relative fold change differences for comparisons of C57BL/6J infected to C57BL/6J control, CBA/Ca infected to CBA/Ca control, CBA/Ca control to C57BL/6J control and C57BL/6J infected to CBA/Ca infected samples.

https://doi.org/10.1371/journal.pntd.0004837.s002 (XLSX)

S1 Table. Exclusively expressed proteins in C57BL/6J infected, C57BL/6J control, CBA/Ca infected and CBA/Ca control mice.

Only proteins exclusive to or absent from all three replicates in a given sample are reported. A zero log₂ normalised intensity value indicates a protein that was absent or undetected in a sample. https://doi.org/10.1371/journal.pntd.0004837.s003 (XLSX)

S2 Table. Fisher exact test for enrichment in Uniprot Keywords GOBP, GCC and KEGG terms.

Fisher exact test with a Benjamini-Hochberg corrected FDR of 2%, on t-test significant differential abundant and exclusively expressed proteins from C57BL/6J and CBA/Ca mice with and without *Ascaris* infection. NTD: No term determined. https://doi.org/10.1371/journal.pntd.0004837.s004 (XLSX)

S1 Fig. Interaction network analysis of up regulated proteins in CBA/Ca and C57BL/6J control and infected mice.

Protein interaction information was obtained from the STRING database using gene lists extracted for statistically significant differentially abundant (SSDA) proteins from pair wise t-tests (p< 0.05). Each node represents a protein and each connecting line represents an interaction, the extent of evidence for which is represented by the width of the line. Statistically enriched KEGG and biological process Gene Ontology (GO) descriptors were examined to identify clusters of proteins enriched within SSDA protein of increased abundance for (A) CBA/Ca control in comparison to C57BL/6J control, (B) C57BL/6J control in comparison to C8A/Ca infected in comparison to C57BL/6J infected and (D) C57BL/6J infected in comparison to CBA/Ca infected mice.

https://doi.org/10.1371/journal.pntd.0004837.s005 (TIF)

S2 Fig. KEGG analysis displaying significantly changing enzymes involved in the process of oxidative phosphorylation and the components of the ribosome in CBA/Ca and C57BL/6J control and infected mice.

(A) Differentially expressed proteins involved in oxidative phosphorylation. CBA/Ca control and infected mice display a higher abundance of enzymes (red, purple and cyan) compared to C57BL/6J control and infected mice (blue, orange and pink). A significant proportion of these proteins have higher abundances in CBA/Ca control mice with respect to their C57BL/6J counterparts, indicating a clear intrinsic difference between the strains. (B) Differentially expressed proteins involved with the ribosome. C57BL/6J control samples had less ribosomal proteins than their CBA/Ca control counterparts (blue and red) indicative of an innate difference between these two strains. Under infection this intrinsic difference is also observed, with higher abundance for ribosomal proteins in CBA/Ca mice (purple and orange) observed. https://doi.org/10.1371/journal.pntd.0004837.s006

(TIF)

S3 Fig. KEGG analysis displaying significantly changing enzymes involved in the TCA Cycle in CBA/Ca and C57BL/6J control and infected mice.

(A) Differentially expressed proteins involved in the TCA cycle in control mice. A clear intrinsic difference is observed between CBA/Ca (red) and C57BL/6J mice (blue). (B) Differentially expressed proteins involved in the TCA cycle in infected mice. Under infection C57BL/6J mice display a higher abundance of TCA cycle proteins (yellow). https://doi.org/10.1371/journal.pntd.0004837.s007 (TIF)

S4 Fig. KEGG analysis displaying significantly changing enzymes involved in glutathione metabolism in CBA/Ca and C57BL/6J control and infected mice.

Differentially expressed proteins involved glutathione metabolism. Glutathione metabolism enzymes are more abundant in C57BL/6J control mice (blue) and compared to their CBA/Ca counterparts indiacting an intrinsic difference between the strains.

Under infection C57BL/6J mice had increased expression of glutathione metabolism proteins (blue and cyan) and increase that was less pronounced in CBA/Ca *Ascaris*-infected mice (red). https://doi.org/10.1371/journal.pntd.0004837.s008 (TIF)

Acknowledgments

We thank Peter Stafford, Senior specialist technician, Karen Loxton, Kate Minogue and Gillian Johnston (Trinity College Dublin) for assistance with the mouse experiments, Peter Nejsum (Faculty of Health and Medical Sciences, The University of Copenhagen, Denmark) for provision of *Ascaris* eggs and Caroline Batchelor (Maynooth University) for assistance with mass spectrometry. We dedicate this manuscript to the memory of our colleague Noel Murphy (Maynooth University), a molecular parasitologist and man of extraordinary kindness and humanity.

Author Contributions

Conceived and designed the experiments: CVH TJC.

Performed the experiments: GD TJC AJRC JCC.

Analyzed the data: GD JCC.

Contributed reagents/materials/analysis tools: CVH JCC.

Wrote the paper: GD TJC AJRC CVH JCC.

References

- 1. Holland CV, (Editor). Ascaris: The Neglected Parasite: Elsevier; 2013.
- Pullan RL, Smith JL, Jasrasaria R, Brooker SJ. Global numbers of infection and disease burden of soil transmitted helminth infections in 2010. Parasit Vectors. 2014;7:37. pmid:24447578
 View Article • PubMed/NCBI • Google Scholar
- 3. Thamsbourg SM, Nejsum P, Mejer H. Chapter 14: Impact of Ascaris suum in livestock. Ascaris: The Neglected Parasite: Elsevier; 2013. p. 363-81.
- 4. O'Lorcain P, Holland CV. The public health importance of Ascaris lumbricoides. Parasitology. 2000;121 Suppl:S51–71. pmid:11386692
 View Article PubMed/NCBI Google Scholar
- Holland CV. Predisposition to ascariasis: patterns, mechanisms and implications. Parasitology. 2009;136(12):1537–47. pmid:19450374 View Article • PubMed/NCBI • Google Scholar
- 6. Crofton HD. A quantitative approach to parasitism. Parasitology. 63:343–64.
 View Article PubMed/NCBI Google Scholar
- 7. Shaw DJ, Dobson P. Patterns of macroparasite abundance and aggregation in wildlife populations: a quantitative review. Parasitology. 1995;111:S111–S33. pmid:8632918
 View Article • PubMed/NCBI • Google Scholar
- 8. Haswell-Elkins MR, Elkins DB, Anderson RM. Evidence for predisposition in humans to infection with Ascaris, hookworm, Enterobius and Trichuris in a South Indian fishing community. Parasitology. 1987;95 (Pt 2):323–37. pmid:3696767
 View Article • PubMed/NCBI • Google Scholar
- 9. Holland CV, Asaolu SO, Crompton DW, Stoddart RC, Macdonald R, Torimiro SE. The epidemiology of Ascaris lumbricoides and other soil-transmitted helminths in primary school children from IIe-Ife, Nigeria. Parasitology. 1989;99 Pt 2:275–85. pmid:2594419
 View Article • PubMed/NCBI • Google Scholar
- Quinnell RJ, Griffin J, Nowell MA, Raiko A, Pritchard DI. Predisposition to hookworm infection in Papua New Guinea. Transactions of the Royal Society of Tropical Medicine and Hygiene. 2001;95(2):139–42. pmid:11355543
 View Article • PubMed/NCBI • Google Scholar
- Boes J, Medley GF, Eriksen L, Roepstorff A, Nansen P. Distribution of *Ascaris suum* in experimentally and naturally infected pigs and comparison with *Ascaris lumbricoides* infections in humans. Parasitology. 1998;117 (Pt 6):589–96. pmid:9881384
 View Article • PubMed/NCBI • Google Scholar
- Medley GF, Hollingsworth D. MDA helminth control: more questions than answers. Lancet Global Health; 2015. p. E583–E4. pmid:26385292 View Article • PubMed/NCBI • Google Scholar
- Holland CV, Boes J. Distributions and predispositions. In: Holland CV, Kennedy M, editors. The Geohelminths: Ascaris, Trichuris and Hookworm. London: Kluwer Academic Press; 2002. p. 1–24.
- 14. Keymer AE, Pagal M. Predisposition to helminth infection. In: Schad GA, Warren KS, editors. Hookworm disease: current status and new directions. London: Taylor and Francis; 1990. p. 177–209.
- 15. Sprent JF. The life cycles of nematodes in the family Ascarididae Blanchard 1896. Journal of Parasitology. 1954;40:608–17. pmid:13202021 View Article • PubMed/NCBI • Google Scholar

- 16. Stephenson LS. Ascariasis. In: Stephenson LS, Holland CV, editors. The impact of Helminth Infection on Human nutrition. London and New York: Taylor and Francis; 1987. p. 89-127.
- 17. Javid G, Wani NA, Gulzar GM, Khan BA, Shah AH, Shah OJ, et al. Ascaris-induced liver abscess. World J Surg. 1999;23(11):1191-4. pmid:10501884 View Article • PubMed/NCBI • Google Scholar
- 18. Sakakibara A, Baba K, Niwa S, Yagi T, Wakayama H, Yoshida K, et al. Visceral larva migrans due to Ascaris suum which presented with eosinophilic pneumonia and multiple intra-hepatic lesions with severe eosinophil infiltration-outbreak in a Japanese area other than Kyushu. Intern Med. 2002;41(7):574-9. pmid:12132528 View Article • PubMed/NCBI • Google Scholar
- 19. Kakihara D, Yoshimitsu K, Ishigami K, Irie H, Aibe H, Tajima T, et al. Liver lesions of visceral larva migrans due to Ascaris suum infection: CT findings. Abdom Imaging. 2004;29(5):598-602. pmid:15383899 View Article • PubMed/NCBI • Google Scholar
- 20. Schwartz B, Alicata JE. Ascaris larvae as a cause of liver and lung lesions in swine. The Journal of Parasitology. 1932;19(1):17-24. View Article • PubMed/NCBI • Google Scholar
- 21. Ronéus O. Studies on the aetiology and pathogenesis of white spots in the liver of pigs. Acta Vet Scand. 1966;7:Suppl 16:1–112. pmid:5949007 View Article • PubMed/NCBI • Google Scholar
- 22. Copeman DB, Gaafar SM. Sequential development of hepatic lesions of ascaridosis in colostrum-deprived pigs. Aust Vet J. 1972;48(5):263-8. pmid:4672116 View Article • PubMed/NCBI • Google Scholar
- 23. Pérez J, García PM, Mozos E, Bautista MJ, Carrasco L. Immunohistochemical characterization of hepatic lesions associated with migrating larvae of Ascaris suum in pigs. J Comp Pathol. 2001;124(2–3):200–6. pmid:11222018 View Article • PubMed/NCBI • Google Scholar
- 24. Frontera E, Roepstorff A, Gázquez A, Reina D, Serrano FJ, Navarrete I. Immunohistochemical distribution of antigens in liver of infected and immunized pigs with Ascaris suum. Vet Parasitol. 2003;111(1):9-18. pmid:12523975 View Article • PubMed/NCBI • Google Scholar
- 25. Loeffler W. Zur Differentialdiagnose der Lungen-infiltrierungen. II Ueber fluchtige Succedanininfiltrate (mit Eosinophilie). Beiträge zur Klinik der Tuberkulose, 1932;79;368-82, View Article • PubMed/NCBI • Google Scholar
- 26. Loeffler W. Transient lung infiltrations with blood eosinophilia. Int Arch Allergy Appl Immunol. 1956;8(1–2):54–9. pmid:13331628 View Article • PubMed/NCBI • Google Scholar
- 27. Ribeiro JD, Fisher GB. Eosinophilic lung diseases. Paediatric Respiratory Reviews. 2002;3:278-84. pmid:12457596 View Article • PubMed/NCBI • Google Scholar
- 28. Taffs LF. Immunological studies on experimental infection of pigs with Ascaris suum Goeze, 1782. VI. The histopathology of the liver and lung. J Helminthol. 1968;42(1):157-72. pmid:5659046 View Article • PubMed/NCBI • Google Scholar
- 29. Holland CV, Behnke JM, Dold C. Chapter 5: Larval Ascariasis: Impact, Significance, and Model Organisms. In: Holland CV, editor. Ascaris: the neglected parasite: Newnes; 2013. p. 108-25.
- 30. Lewis R, Behnke JM, Stafford P, Holland CV. The development of a mouse model to explore resistance and susceptibility to early Ascaris suum infection. Parasitology. 2006;132(Pt 2):289-300. pmid:16209722 View Article • PubMed/NCBI • Google Scholar
- 31. Lewis R, Behnke JM, Cassidy JP, Stafford P, Murray N, Holland CV. The migration of Ascaris suum larvae, and the associated pulmonary inflammatory response in susceptible C57BL/6j and resistant CBA/Ca mice. Parasitology. 2007;134(Pt 9):1301-14. pmid:17381887 View Article • PubMed/NCBI • Google Scholar
- 32. Dold C, Cassidy JP, Stafford P, Behnke JM, Holland CV. Genetic influence on the kinetics and associated pathology of the early stage (intestinal-hepatic) migration of Ascaris suum in mice. Parasitology. 2010;137(1):173-85. 3. pmid:19765333 View Article • PubMed/NCBI • Google Scholar
- 33. Simons JE, Gray CA, Lawrence RA. Absence of regulatory IL-10 enhances innate protection against filarial parasites by a neutrophil-independent mechanism. Parasite Immunol. 2010;32(7):473-8. pmid:20591117 View Article • PubMed/NCBI • Google Scholar
- 34. Lemos de Souza V, Ascenção Souza J, Correia Silva TM, Sampaio Tavares Veras P, Rodrigues de-Freitas LA. Different Leishmania species determine distinct profiles of immune and histopathological responses in CBA mice. Microbes Infect. 2000;2(15):1807–15. pmid:11165924 View Article • PubMed/NCBI • Google Scholar
- 35. Jones DE, Ackermann MR, Wille U, Hunter CA, Scott P. Early enhanced Th1 response after Leishmania amazonensis infection of C57BL/6 interleukin-10-deficient mice does not lead to resolution of infection. Infect Immun. 2002;70(4):2151-8. pmid:11895981 View Article • PubMed/NCBI • Google Scholar

- 36. Hirata M, Hara T, Kage M, Fukuma T, Sendo F. Neutropenia augments experimentally induced Schistosoma japonicum egg granuloma formation in CBA mice, but not in C57BL/6 mice. Parasite Immunol. 2002;24(9–10):479–88. pmid:12654090
 View Article PubMed/NCBI Google Scholar
- Salazar-Castañon VH, Legorreta-Herrera M, Rodriguez-Sosa M. Helminth parasites alter protection against *Plasmodium* infection. Biomed Res Int. 2014;2014:913696. pmid:25276830
 View Article PubMed/NCBI Google Scholar
- Bantscheff M, Lemeer S, Savitski MM, Kuster B. Quantitative mass spectrometry in proteomics: critical review update from 2007 to the present. Anal Bioanal Chem. 2012;404(4):939–65. pmid:22772140
 View Article PubMed/NCBI Google Scholar
- 39. Saka HA, Thompson JW, Chen YS, Kumar Y, Dubois LG, Moseley MA, et al. Quantitative proteomics reveals metabolic and pathogenic properties of *Chlamydia trachomatis* developmental forms. Mol Microbiol. 2011;82(5):1185–203. pmid:22014092
 View Article PubMed/NCBI Google Scholar
- 40. Chopra S, Ramkissoon K, Anderson DC. A systematic quantitative proteomic examination of multidrug resistance in *Acinetobacter baumannii*. J Proteomics. 2013;84:17–39. pmid:23542354
 View Article • PubMed/NCBI • Google Scholar
- Panichakul T, Ponnikorn S, Roytrakul S, Paemanee A, Kittisenachai S, Hongeng S, et al. *Plasmodium vivax* inhibits erythroid cell growth through altered phosphorylation of the cytoskeletal protein ezrin. Malar J. 2015;14:138. pmid:25889165
 View Article PubMed/NCBI Google Scholar
- 42. Minogue K. A study comparing hepatic inflammation of C57BL/6J and CBA/Ca male mice infected with Ascaris suum when exposed and unexposed to female mice presence. Moderatorship Thesis: Trinity College Dublin; 2014.
- Hubner NC, Bird AW, Cox J, Splettstoesser B, Bandilla P, Poser I, et al. Quantitative proteomics combined with BAC TransgeneOmics reveals in vivo protein interactions. J Cell Biol. 2010;189(4):739–54. pmid:20479470
 View Article PubMed/NCBI Google Scholar
- Cox J, Neuhauser N, Michalski A, Scheltema RA, Olsen JV, Mann M. Andromeda: a peptide search engine integrated into the MaxQuant environment. J Proteome Res. 2011;10(4):1794–805. pmid:21254760
 View Article • PubMed/NCBI • Google Scholar
- 45. UniProt Consortium. UniProt: a hub for protein information. Nucleic Acids Res. 2015;43(Database issue):D204–12. pmid:25348405 View Article • PubMed/NCBI • Google Scholar
- 46. Cox J, Hein MY, Luber CA, Paron I, Nagaraj N, Mann M. MaxLFQ allows accurate proteome-wide label-free quantification by delayed normalization and maximal peptide ratio extraction. Molecular & Cellular Proteomics. 2014.
 View Article • PubMed/NCBI • Google Scholar
- 47. Quackenbush J. Microarray data normalization and transformation. Nat Genet. 2002;32 Suppl:496–501. pmid:12454644
 View Article PubMed/NCBI Google Scholar
- 48. Deeb SJ, D'Souza RC, Cox J, Schmidt-Supprian M, Mann M. Super-SILAC allows classification of diffuse large B-cell lymphoma subtypes by their protein expression profiles. Mol Cell Proteomics. 2012;11(5):77–89. pmid:22442255
 View Article PubMed/NCBI Google Scholar
- 49. Jensen LJ, Kuhn M, Stark M, Chaffron S, Creevey C, Muller J, et al. STRING 8—a global view on proteins and their functional interactions in 630 organisms. Nucleic Acids Res. 2009;37(Database issue):D412–6. pmid:18940858
 View Article PubMed/NCBI Google Scholar
- 50. Kanehisa M, Goto S, Sato Y, Kawashima M, Furumichi M, Tanabe M. Data, information, knowledge and principle: back to metabolism in KEGG. Nucleic Acids Res. 2014;42(Database issue):D199–205. pmid:24214961
 View Article PubMed/NCBI Google Scholar
- 51. Kanehisa M, Goto S. KEGG: kyoto encyclopedia of genes and genomes. Nucleic Acids Res. 2000;28(1):27–30. pmid:10592173 View Article • PubMed/NCBI • Google Scholar
- 52. Côté RG, Griss J, Dianes JA, Wang R, Wright JC, van den Toorn HW, et al. The PRoteomics IDEntification (PRIDE) Converter 2 framework: an improved suite of tools to facilitate data submission to the PRIDE database and the ProteomeXchange consortium. Mol Cell Proteomics. 2012;11(12):1682–9. pmid:22949509
 View Article PubMed/NCBI Google Scholar
- 53. Utzinger J, Raso G, Brooker S, De Savigny D, Tanner M, Ornbjerg N, et al. Schistosomiasis and neglected tropical diseases: towards integrated and sustainable control and a word of caution. Parasitology. 2009;136(13):1859–74. pmid:19906318
 View Article PubMed/NCBI Google Scholar
- 54. Jia TW, Melville S, Utzinger J, King CH, Zhou XN. Soil-transmitted helminth reinfection after drug treatment: a systematic review and meta-analysis. PLoS Negl Trop Dis. 2012;6(5):e1621. pmid:22590656
 View Article PubMed/NCBI Google Scholar

- 55. Dold C, Holland CV. Investigating the underlying mechanism of resistance to Ascaris infection. Microbes Infect. 2011;13(7):624–31. pmid:20934532 View Article • PubMed/NCBI • Google Scholar
- 56. Adam-Vizi V. Production of reactive oxygen species in brain mitochondria: contribution by electron transport chain and non-electron transport chain sources. Antioxid Redox Signal. 2005;7(9-10):1140-9. pmid:16115017 View Article • PubMed/NCBI • Google Scholar
- 57. Kowaltowski AJ, de Souza-Pinto NC, Castilho RF, Vercesi AE. Mitochondria and reactive oxygen species. Free Radic Biol Med. 2009;47(4):333-43. pmid:19427899 View Article • PubMed/NCBI • Google Scholar
- 58. Yang Y, Bazhin AV, Werner J, Karakhanova S. Reactive oxygen species in the immune system. Int Rev Immunol. 2013;32(3):249–70. pmid:23617726 View Article • PubMed/NCBI • Google Scholar
- 59. Pecaut MJ, Gridley DS. The impact of mouse strain on iron ion radio-immune response of leukocyte populations. Int J Radiat Biol. 2010;86(5):409-19. pmid:20397846 View Article • PubMed/NCBI • Google Scholar
- 60. Staecker H, Zheng QY, Van De Water TR. Oxidative stress in aging in the C57B16/J mouse cochlea. Acta Otolaryngol. 2001;121(6):666–72. pmid:11678164 View Article • PubMed/NCBI • Google Scholar
- 61. Johnson KR, Erway LC, Cook SA, Willott JF, Zheng QY. A major gene affecting age-related hearing loss in C57BL/6J mice. Hear Res. 1997;114(1-2):83-92. pmid:9447922 View Article • PubMed/NCBI • Google Scholar
- 62. Distelmaier F, Koopman WJ, van den Heuvel LP, Rodenburg RJ, Mayatepek E, Willems PH, et al. Mitochondrial complex I deficiency: from organelle dysfunction to clinical disease. Brain. 2009;132(Pt 4):833-42. pmid:19336460 View Article • PubMed/NCBI • Google Scholar
- 63. Finsterer J. Leigh and Leigh-like syndrome in children and adults. Pediatr Neurol. 2008;39(4):223–35. pmid:18805359 View Article • PubMed/NCBI • Google Scholar
- 64. Campbell NA, Reece JB, Urry LA, Cain ML, Wasserman SA, Minorsky PV, et al. Biology: A global appoach. England: Pearson; 2014.
- 65. de Macedo Soares MF, de Macedo MS. Modulation of anaphylaxis by helminth-derived products in animal models. Curr Allergy Asthma Rep. 2007;7(1):56-61. pmid:17504662 View Article • PubMed/NCBI • Google Scholar
- 66. Soares MF, Araújo CA. Helminth products as a potential therapeutic strategy for inflammatory diseases. Inflamm Allergy Drug Targets. 2008;7(2):113-8. pmid:18691141 View Article • PubMed/NCBI • Google Scholar
- 67. Maizels RM. Yazdanbakhsh M. Immune regulation by helminth parasites; cellular and molecular mechanisms. Nat Rev Immunol. 2003;3(9):733-44. pmid:12949497 View Article • PubMed/NCBI • Google Scholar
- 68. Jex AR, Liu S, Li B, Young ND, Hall RS, Li Y, et al. Ascaris suum draft genome. Nature. 2011;479(7374):529–33. pmid:22031327 View Article • PubMed/NCBI • Google Scholar
- 69. Wang T, Van Steendam K, Dhaenens M, Vlaminck J, Deforce D, Jex AR, et al. Proteomic analysis of the excretory-secretory products from larval stages of Ascaris suum reveals high abundance of glycosyl hydrolases. PLoS Negl Trop Dis. 2013;7(10):e2467. pmid:24098821 View Article • PubMed/NCBI • Google Scholar
- 70. Donnelly S, Stack CM, O'Neill SM, Sayed AA, Williams DL, Dalton JP. Helminth 2-Cys peroxiredoxin drives Th2 responses through a mechanism involving alternatively activated macrophages. FASEB J. 2008;22(11):4022-32. pmid:18708590 View Article • PubMed/NCBI • Google Scholar
- 71. Wen JJ, Garg NJ. Mitochondrial complex III defects contribute to inefficient respiration and ATP synthesis in the myocardium of Trypanosoma cruziinfected mice. Antioxid Redox Signal. 2010;12(1):27-37. pmid:19624257 View Article • PubMed/NCBI • Google Scholar
- 72. Wen JJ, Garg NJ. Mitochondrial generation of reactive oxygen species is enhanced at the Q(o) site of the complex III in the myocardium of Trypanosoma cruzi-infected mice: beneficial effects of an antioxidant. J Bioenerg Biomembr. 2008;40(6):587-98. pmid:19009337 View Article • PubMed/NCBI • Google Scholar
- 73. Paiva CN, Feijó DF, Dutra FF, Carneiro VC, Freitas GB, Alves LS, et al. Oxidative stress fuels Trypanosoma cruzi infection in mice. J Clin Invest. 2012;122(7):2531-42. pmid:22728935 View Article • PubMed/NCBI • Google Scholar
- 74. Smith NC, Bryant C. The effect of antioxidants on the rejection of Nippostrongylus brasiliensis. Parasite Immunol. 1989;11(2):161–7. pmid:2717198

View Article • PubMed/NCBI • Google Scholar

- 75. Bass DA, Szejda P. Mechanisms of killing of newborn larvae of Trichinella spiralis by neutrophils and eosinophils. Killing by generators of hydrogen peroxide in vitro. J Clin Invest. 1979;64(6):1558-64. pmid:41002 View Article • PubMed/NCBI • Google Scholar
- 76. Henkle-Dührsen K, Kampkötter A. Antioxidant enzyme families in parasitic nematodes. Mol Biochem Parasitol. 2001;114(2):129-42. pmid:11378193 View Article • PubMed/NCBI • Google Scholar
- 77. Eckelt VH, Liebau E, Walter RD, Henkle-Dührsen K. Primary sequence and activity analyses of a catalase from Ascaris suum. Mol Biochem Parasitol. 1998;95(2):203-14. pmid:9803413 View Article • PubMed/NCBI • Google Scholar
- 78. Tsuji N, Kasuga-Aoki H, Isobe T, Yoshihara S. Cloning and characterisation of a peroxiredoxin from the swine roundworm Ascaris suum. Int J Parasitol. 2000;30(2):125-8. pmid:10704594 View Article • PubMed/NCBI • Google Scholar
- 79. Marí M, Morales A, Colell A, García-Ruiz C, Kaplowitz N, Fernández-Checa JC. Mitochondrial glutathione: features, regulation and role in disease. Biochim Biophys Acta. 2013;1830(5):3317-28. pmid:23123815 View Article • PubMed/NCBI • Google Scholar
- 80. Venditti P, Di Stefano L, Di Meo S. Mitochondrial metabolism of reactive oxygen species. Mitochondrion. 2013;13(2):71-82. pmid:23376030 View Article • PubMed/NCBI • Google Scholar
- 81. Masure D, Vlaminck J, Wang T, Chiers K, Van den Broeck W, Vercruysse J, et al. A role for eosinophils in the intestinal immunity against infective Ascaris suum larvae. PLoS Negl Trop Dis. 2013;7(3):e2138. pmid:23556022 View Article • PubMed/NCBI • Google Scholar
- 82. Malik A, Batra JK. Antimicrobial activity of human eosinophil granule proteins: involvement in host defence against pathogens. Crit Rev Microbiol. 2012;38(2):168-81. pmid:22239733 View Article • PubMed/NCBI • Google Scholar
- 83. Jenne CN, Kubes P. Immune surveillance by the liver. Nat Immunol. 2013;14(10):996–1006. pmid:24048121 View Article • PubMed/NCBI • Google Scholar
- 84. Bertolino P, Bowen DG. Malaria and the liver: immunological hide-and-seek or subversion of immunity from within? Front Microbiol. 2015;6:41. pmid:25741320 View Article • PubMed/NCBI • Google Scholar
- 85. Murphy K. Janeway's immunobiology. 8th ed: Garland Science, Taylor & Francis Group, LLc; 2012.
- 86. Sarma JV, Ward PA. The complement system. Cell Tissue Res. 2011;343(1):227-35. pmid:20838815 View Article • PubMed/NCBI • Google Scholar
- 87. Maizels RM, Philipp M, Ogilvie BM. Molecules on the surface of parasitic nematodes as probes of the immune response in infection. Immunol Rev. 1982;61:109-36. pmid:6174410 View Article • PubMed/NCBI • Google Scholar
- 88. Rees-Roberts D, Mullen LM, Gounaris K, Selkirk ME. Inactivation of the complement anaphylatoxin C5a by secreted products of parasitic nematodes. Int J Parasitol. 2010;40(5):527-32. pmid:19874826 View Article • PubMed/NCBI • Google Scholar
- 89. Marenholz I, Heizmann CW, Fritz G. S100 proteins in mouse and man: from evolution to function and pathology (including an update of the nomenclature). Biochem Biophys Res Commun. 2004;322(4):1111-22. pmid:15336958 View Article • PubMed/NCBI • Google Scholar
- 90. Ehrchen JM, Sunderkötter C, Foell D, Vogl T, Roth J. The endogenous Toll-like receptor 4 agonist S100A8/S100A9 (calprotectin) as innate amplifier of infection, autoimmunity, and cancer. J Leukoc Biol. 2009;86(3):557-66. pmid:19451397 View Article • PubMed/NCBI • Google Scholar
- 91. Contreras I, Shio MT, Cesaro A, Tessier PA, Olivier M. Impact of neutrophil-secreted myeloid related proteins 8 and 14 (MRP 8/14) on leishmaniasis progression. PLoS Negl Trop Dis. 2013;7(9):e2461. pmid:24086787 View Article • PubMed/NCBI • Google Scholar
- 92. Vogl T, Tenbrock K, Ludwig S, Leukert N, Ehrhardt C, van Zoelen MA, et al. Mrp8 and Mrp14 are endogenous activators of Toll-like receptor 4, promoting lethal, endotoxin-induced shock. Nat Med. 2007;13(9):1042-9. pmid:17767165 View Article • PubMed/NCBI • Google Scholar
- 93. Srivastava P. Roles of heat-shock proteins in innate and adaptive immunity. Nat Rev Immunol. 2002;2(3):185–94. pmid:11913069 View Article • PubMed/NCBI • Google Scholar
- 94. Ryckman C, Vandal K, Rouleau P, Talbot M, Tessier PA. Proinflammatory activities of S100: proteins S100A8, S100A9, and S100A8/A9 induce neutrophil

chemotaxis and adhesion. J Immunol. 2003;170(6):3233-42. pmid:12626582 View Article • PubMed/NCBI • Google Scholar

- 95. Edgeworth JD, Abiose A, Jones BR. An immunohistochemical analysis of onchocercal nodules: evidence for an interaction between macrophage MRP8/MRP14 and adult Onchocerca volvulus. Clin Exp Immunol. 1993;92(1):84-92. pmid:8467568 View Article • PubMed/NCBI • Google Scholar
- 96. Mohr I, Sonenberg N. Host translation at the nexus of infection and immunity. Cell Host Microbe. 2012;12(4):470-83. pmid:23084916 View Article • PubMed/NCBI • Google Scholar
- 97. Fontana MF, Banga S, Barry KC, Shen X, Tan Y, Luo ZQ, et al. Secreted bacterial effectors that inhibit host protein synthesis are critical for induction of the innate immune response to virulent Legionella pneumophila. PLoS Pathog. 2011;7(2):e1001289. pmid:21390206 View Article • PubMed/NCBI • Google Scholar
- 98. Jaramillo M, Gomez MA, Larsson O, Shio MT, Topisirovic I, Contreras I, et al. Leishmania repression of host translation through mTOR cleavage is required for parasite survival and infection. Cell Host Microbe. 2011;9(4):331-41. pmid:21501832 View Article • PubMed/NCBI • Google Scholar
- 99. Linder MC. Mobilization of stored iron in mammals: a review. Nutrients. 2013;5(10):4022–50. pmid:24152745 View Article • PubMed/NCBI • Google Scholar
- 100. Hempe JM, Ory-Ascani J, Hsia D. Genetic variation in mouse beta globin cysteine content modifies glutathione metabolism: implications for the use of mouse models. Exp Biol Med (Maywood). 2007;232(3):437-44. View Article • PubMed/NCBI • Google Scholar
- 101.Seller MJ. Experimental red cell chimerism in the heterozygote (Wvw) of the W-series mutants in the house mouse. J Med Genet. 1967;4(1):33-5. pmid:5340572 View Article • PubMed/NCBI • Google Scholar
- 102. Harleman JH. The haemoglobin types of mice. Lab Anim. 1977;11(2):105-8. pmid:865067 View Article • PubMed/NCBI • Google Scholar
- 103.Fortin A, Stevenson MM, Gros P. Susceptibility to malaria as a complex trait: big pressure from a tiny creature. Hum Mol Genet. 2002;11(20):2469–78. pmid:12351583 View Article • PubMed/NCBI • Google Scholar